Andrzej S Tarnawski  
Editor-in-Chief, World Journal of Gastroenterology  

Manuscript NO: 87006  

Re: Trans-anal endoscopic microsurgery for non-adenomatous rectal lesions  

We are very appreciative of the positive and constructive comments from the reviewers. In the following pages, we address issues raised by the reviewers (quoted in bold), followed by our responses. Each response is followed by a description of the changes made and their section in the manuscript. Finally, changes in the manuscript are tracked. We hope the revised manuscript will be found suitable for publication. Please contact us if there are additional comments or questions. We look forward to your response.  

Sincerely,  

Dafna (Shilo) Yaacobi, MD  
Department of plastic surgery & burns  
Rabin Medical Center  
39 Jabutisnki St., Petah Tikva, 4941492, Israel.  
Tel. +972-39376378  
Fax. +972-39219773  
Email: dafna.yaacobi@icloud.com
Response to Reviewers' Comments

Reviewer #1:
Hope you will come with more numbers in coming years..

Dear Reviewer,

Thank you, we also do.

Reviewer #2:
This study investigates the safety and efficacy of trans-anal endoscopic microsurgery for rare and non-adenomatous rectal and retro-rectal lesions, emphasizing that TEM should be reasonably considered as an important surgical method in the treatment of rare lesions. Further studies require larger cohorts and prospective design to validate these findings.

Dear Reviewer,

Thank you, we will definitely continue in order to validate our results.

Sincerely,

Dafna Shilo Yaacobi, MD
Department of plastic surgery & burns
Rabin Medical Center
39 Jabutisnki St., Petah Tikva, 4941492, Israel.
Tel. +972-39376378 Fax. +972-39219773
Email: dafna.yaacobi@icloud.com